Contents lists available at ScienceDirect



Environmental Innovation and Societal Transitions

journal homepage: www.elsevier.com/locate/eist



Exploring financiers' beliefs and behaviours at the outset of low-carbon transitions: A shipping case study



Marie Fricaudet^{*}, Sophia Parker, Nishatabbas Rehmatulla

UCL Energy Institute, 14 Upper Woburn Place, London WC1H ONN, UK

ARTICLE INFO

Keywords: Shipping finance Socio-technical transitions Climate mitigation

ABSTRACT

Ship financiers play an important role in the shipping industry's transition to a low carbon economy as they shape the financing terms provided to shipowners and the type of assets that are financed, but their transition-related beliefs are unclear. This paper proposes a novel theoretical framework which categorises financiers during low-carbon transitions into five archetypes. It also provides novel empirical insights on the beliefs and ambitions of shipping financiers regarding their future role and the risks to their assets using semi-structured interviews with ten financiers and two shipowners. The results show that the majority of interviewees expressed an ambition to support incumbent shipowners in the transition, but expressed the need for regulation to ensure financiers properly measure climate risks (e.g. the EU taxonomy and differentiated capital requirements for green and brown assets). This highlights the potential, but also the limits, of voluntary initiatives from financiers to promote green investments.

1. Introduction

Shipping represents about 3 % of the world's greenhouse gas emissions, and emissions are projected to increase by between 90–130 % of 2008 emissions by 2050 for a range of long-term economic and energy scenarios (Faber et al., 2020) if it does not rapidly increase the energy efficiency of the fleet and adopt zero/low-carbon fuels. Several technological and operational measures already exist to increase the energy efficiency of ships, such as speed reduction or technology (e.g., wind assistance, air lubrification) but they need to be combined with zero/low-carbon fuels to be aligned with a 1.5° pathway (Smith et al., 2023). Because ships are long-lived assets (about 25 years), most ships ordered this decade will have to shift to alternative fuels over their lifetime or be scrapped early for shipping to remain in line with a 1.5° pathway. However, the zero/low-carbon marine fuel market is nascent with many different candidate zero/low-carbon fuels. For example, hydrogen-based e-fuels such as ammonia, viewed as the most economical option, are still in demo/pilot phase (Scarbrough et al., 2022). The immature state of the marine fuel market and the numerous possible low/zero-carbon fuel pathways creates an uncertain investment environment.

The majority of ship finance historically comes from shipowners' private equity and debt provided by commercial banks, although a growing share comes from non-banking sources (e.g., leasing, alternative lending, private equity from funds, public markets and investors, Drobetz et al. 2013; Del Gaudio 2018). The shipping debt market is concentrated, with the top 10 shipping financiers representing around 45% of the banks' shipping debt portfolio (calculated from Petropoulos, 2021). This implies that a small number of financiers dominate financing decisions. Ship financiers shape the financing terms provided to shipowners and the type of assets that

* Corresponding author.

E-mail address: m.fricaudet@ucl.ac.uk (M. Fricaudet).

https://doi.org/10.1016/j.eist.2023.100788

Received 10 May 2023; Received in revised form 16 October 2023; Accepted 7 November 2023

Available online 18 November 2023

^{2210-4224/© 2023} The Authors. Published by Elsevier B.V. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

are financed. More recently, they are facing new risk factors as a result of climate change. These risks stem from both demand-side climate risk - the destruction of seaborne fossil-fuel cargo as land-based sectors decarbonise - and supply-side climate risks from the transition of ships propelled by fossil fuels to alternative low/zerocarbon fuels (Bullock et al., 2020). Demand-side and supply-side climate risks might lead to an unexpected devaluation of the ship assets due to changes in regulation, technology, or consumer demand which has a cascading impact on a shipping financier's profitability. For example, in a conventional bank loan, the financier would assume ownership of the asset in the event that the shipowner defaults and would recover the amount that the ship is valued at the time of default. There is a potential gap in financing the transition to low/zero-carbon shipping due to traditional banks reducing their exposure from the sector following the 2007-2008 financial crisis (Global Financial Crisis) (Gong et al., 2013; Schinas and Bergmann, 2021; Schinas et al., 2018). Financiers' decisions have an impact not only on their profitability but also the ability to transition to low/zero-carbon ships in the future. For example, funding fossil fueled ships could create technology lock-in and make the transition more difficult to low/zero-carbon shipping. In 2019, a group of shipping financiers recognised their role within shipping finance to promote responsible environmental practices and formed the Poseidon Principles, a framework for including climate considerations in shipping finance for assessing and disclosing the climate alignment of ship finance portfolios. The signatory members have grown from 10 founding members to 34 leading banks, representing approximately \$200 billion in shipping finance. While there is a growing but conflicting set of evidence on the pricing of climate risks in financial instruments from the econometric literature (Bingler, 2022; Degryse et al., 2021; Delis et al., 2019; Fatica et al., 2021; Hachenberg and Schiereck, 2018; Seltzer et al., 2022), there is little empirical evidence about the role of shipping financiers in sustainability transitions (e.g., how they view climate risk, the nature and drivers of their financing behaviour, and how they are adapting their tools and instruments). Second, while the Multi-level Perspective (MLP) can be used as a theoretical framework to describe an energy transition and the role of different actors, it characterises the role of financiers as functional, resource-providing and lacking agency (Naidoo, 2020).

The paper therefore aims to fill these gaps by firstly characterising financiers' behaviours in a conceptual framework extending the MLP and building on theoretical insights from the Adaptive Market Hypothesis to describe how financiers adapt their expectations following a breakdown in heuristics. Second, it aims to elicit financiers' transition-related beliefs through qualitative social research methods such as interviews in order to understand the role of shipping financiers in sustainability transitions.

The remainder of the article is structured as follows: Section 2 discusses the transition literature with a focus on the multi-level perspective and the finance industry's role in transitions, drawing from other industries like electric utilities. Section 2.4 provides the extended theoretical framework to describe financiers into 5 archetypes. Section 3 provides the context and methodological approach of the study, while Section 4 presents the results and discussion. Section 5 concludes.

2. Literature review and theoretical background

2.1. Finance in shipping

Because of the reliance of the shipping industry on high-value assets, accessing substantial capital to replace an ageing fleet and funding the second-hand ship market is essential (Alexandridis et al., 2018). The longevity of shipping companies has been historically closely tied to their ability to secure financing with favourable terms, particularly low interest rates (Stopford, 2009). Pangalos (2023) argues that because larger shipowners have stronger bargaining power with fuel providers and clients, higher technical management capacity, and better access to sources of capital, they might be better placed to acquire low/zero-carbon ships. The popularity of bank borrowings among ship-owning companies can be attributed to several factors listed by Alexandridis et al. (2018):

- Lower cost and accessibility.
- Preservation of ownership structure, which is often family-oriented and highly concentrated.
- Confidentiality: raising funds through bank loans does not necessitate public disclosure of strategic, financial, and operational information, unlike methods like IPOs and corporate bond issues (Kavussanos and Tsouknidis, 2014, 2016).
- Relationship banking: historically, shipping bank loans have been granted based on relationship banking principles. This fosters long-term rapport, amicable trust, and information sharing between the borrowing company and the bank (Alexandridis et al., 2018; Gavalas and Syriopoulos, 2015; Mitroussi et al., 2016).

Banks limited their exposure to the shipping industry due to weak market conditions and stricter BASEL III capital requirements imposed by the Basel Committee on Banking Supervision following the Global Financial Crisis (Gong et al., 2013). Despite this trend, lending remains the main source of external shipping finance (Drobetz et al., 2013; Del Gaudio, 2018) and the literature on alternative sources of capital is limited (Alexandridis et al., 2018).

The existing literature on shipping finance covers a large range of topics including the sources of finance and capital structure, valuation methods and risk management (an in-depth review of shipping finance is available in Alexandridis et al. 2018). However, the focus of this literature has been on shipowners rather than the sources of finance itself. Even then, the focus has been on a default risk assessment of financial instruments (see Alexandridis et al. 2018 for a review), which does not consider the agency of the finance providers. Exceptions include Gong et al. (2013)'s survey on banks from the shipping sector after 2009 and Mitchell and Rehmatulla (2016)'s interviews with debt and equity providers on the topic of energy efficiency and stranded assets, and Gavalas and Syriopoulos (2015), Lee and Pak (2018)'s review of banks' decision-making drivers.

2.2. Transitions studies and the multi-level perspective

This article builds on the multi-level perspective (MLP) framework, derived from transition theories. The MLP explains the uptake of technological innovations in markets that are associated with key societal and economic functions such as energy or transport (Geels, 2002; Geels and Schot, 2007; Markard et al., 2012; Markard and Truffer, 2008). According to the MLP framework, there are three layers that interact: exogenous landscape pressure, the socio-technical regime, and niches Fig. 1. Exogenous landscape pressure represents a change in civil society's awareness about an issue (e.g. environmental) which exerts pressure on the socio-technical regime (e.g. incumbent industry) (Geels, 2002; Geels and Schot, 2007; Markard et al., 2012; Markard and Truffer, 2008). Socio-technical regimes are stable economic and social configurations, however they can be destabilized if exogenous pressure, such as the growing public pressure to decarbonize industry sectors including the shipping industry, create opportunities for niche innovations to break through (Geels, 2002; Geels and Schot, 2007; Markard et al., 2012; Markard and Truffer, 2008). The third layer are niches which develop in protected spaces and can provide innovations. If these niches are successful, they are taken up by incumbents and can replace the previous socio-technical regime (Geels, 2002; Geels, 2002; Geels and Schot, 2007; Markard et al., 2012; Markard et al., 2012; Markard and Truffer, 2008).

Pettit et al. (2018) and Stalmokaite and Yliskylä-Peuralahti (2019) both use the MLP framework to describe the shipping industry. They describe the shipping regime as comprised of a large range of industry incumbents including shipowners, charterers and customers, ports and fuel providers and regulators (see Fig. 1) (Pettit et al., 2018; Stalmokaite and Yliskylä-Peuralahti, 2019). Shipping case studies include electric or hydrogen-fuelled coastal shipping in Norway (Bach et al., 2020, 2021; Bergek et al., 2021).

Socio-technical transitions can evolve differently depending on the nature of the innovations and landscape pressures (Geels et al.,



Fig. 1. MLP of shipping low/zero-carbon transitions (Stalmokaite and Yliskylä-Peuralahti, 2019).

Table 1

Name	Type of financier	Location of headquarter	Signatory to Poseidon Principles	Size of shipping portfolio
Interview 1	Commercial bank	North America	Yes	\$5–10bn
Interview 2	Commercial bank	Western Europe	Yes	>\$10bn
Interview 3	Alternative financier	Western Europe	No	\$0–5bn
Interview 4	Commercial bank	Western Europe	Yes	>\$10bn
Interview 5	Commercial bank	Western Europe	No	>\$10bn
Interview 6	Asset manager	Western Europe	No	\$0–5bn
Interview 7	Commercial bank	Asia	Yes	>\$10bn
Interview 8	Shipowner	Western Europe	No	\$0–5bn
Interview 9	Commercial bank	Western Europe	Yes	\$5–10bn
Interview 10	Shipowner	Western Europe	No	\$0–5bn
Interview 11	Commercial bank	Asian branch of a North American bank	Yes	\$5–10bn
Interview 13	Commercial bank	Western Europe	Yes	\$5–10bn

2016; Geels and Schot, 2007). In some cases, incumbents are replaced by niche entrants who introduce *competitive* innovations, i.e. innovations which aim at replacing the existing regime technology (Geels and Schot, 2007) (e.g. the uptake of small-scale renewables in Germany by entrants such as citizens, cooperatives and farmers, Geels et al. 2016). In others, *symbiotic* innovations, i.e. innovations that enhance the current system technology by enabling it to address issues and improve its performance (Geels and Schot, 2007), are adopted by regime incumbents who then adapt to the new socio-technical regime (e.g. the uptake of large-scale onshore and offshore wind by incumbent utilities in the UK) (Geels et al., 2016). Baresic (2020) and Pettit et al. (2018) suggest that because shipping assets are capital intensive and have long lifespans, the industry is conservative to radical innovations and has a strong path dependency. Baresic (2020), who bases his argument on the transition to LNG as a marine fuel, argues that the involvement of industry incumbents, in particular shipowners, and the development of symbiotic innovations (e.g. dual-fuel engines or drop-in fuels) are likely to be necessary conditions for a successful shipping transition to take place.

2.3. Finance in transitions studies

While the transition literature has focused mainly on the behaviour of industry incumbents such as utilities or shipowners, there is increasing empirical evidence on the behaviour of financiers during transitions (Bolton and Foxon, 2015; Bolton al., 2015; Falcone et al., 2018; Geddes et al., 2020, 2018; Geddes and Schmidt, 2020; Hall et al., 2016, 2017; Hughes and Downie, 2021; Monk and Perkins, 2020; Seyfang and Gilbert-Squires, 2019; Urban and Wójcik, 2019; Yip and Bocken, 2018; Zhang, 2020). This empirical evidence is reviewed here to inform the theoretical framework used in this article for understanding the behaviour of shipping financiers. Some financiers are found, like most industry incumbents, to ignore the risks and opportunities arising from the growing transition and therefore continue financing the incumbent technology before they realize the threat posed by new industry technology. The Royal Bank of Canada or the China Construction Bank continue to finance the extraction of fossil fuels (Urban and Wójcik, 2019), while Japanese bilateral banks continue to finance coal power plants in South East Asia because they perceive the stranding of their assets unlikely and do not include climate risks assessments (Hughes and Downie, 2021). Most German banks and institutional investors originally, and UK banks until today, were reluctant to finance small-scale renewables projects due to lack of knowledge of the technology, inadequate existing financial tools and instruments for the size and type of project, and perceived riskiness of the projects (Geddes et al., 2018). In some cases, not only are financiers unaware of the upcoming socio-technical transition, but they encourage their clients to grow their incumbent technologies. For example, (Ferguson-Cradler, 2022; Kungl, 2018; Kungl and Geels, 2018) document how German power utilities' shareholders exerted pressures from 2005 to 2010 on company management to ambitiously invest in carbon intensive assets and utilities; and in some cases prevented the shift away from coal to protect their regional coal industry (Geels et al., 2016).

In many cases however, financiers are found to support the transition when they realize the threat and/or opportunity brought by the technology innovation. While this support might be a driver of the uptake of the new technology, we do not suggest that it is the only one nor that it is sufficient to drive the transition alone. In particular, the design of government support – including through the UK and German State banks GIB and KfW - has also played a large role in shaping the transition to large-scale or small-scale decentralised renewables in the UK and Germany, respectively (Geddes et al., 2018; Geels et al., 2016). Because regime industry incumbents and financiers share path-dependent beliefs and the financial regime is aligned to industry incumbents, it could be assumed that regime financiers would be more willing to finance industry incumbents rather than niche entrants. An example of this behaviour is the early support of commercial banks in the development of combined cycle power plants by UK electricity utilities (Bolton and Foxon, 2015; Kern, 2012) and of large-scale onshore and offshore wind projects since the 1990s (Bolton and Foxon, 2015; Geddes et al., 2018), both of which are symbiotic innovations. Commercial banks have however refrained from investing in small-scale renewables in the UK promoted by small niche entrants, due to the perceived risk of the investment and the inadequacy of their financing instruments to those actors (Geddes et al., 2018; Stenzel and Frenzel, 2008).

However, there are also examples in the literature of financiers supporting industry niche entrants. This suggests that regime financiers are not intrinsically loyal to industry incumbents. Examples include the support of the UK Green Investment Bank (GIB) which was formerly owned by the UK government, and later on commercial banks for the uptake of bioenergy in the UK (Geddes et al., 2018). In Germany there was support from KfW, and later on the German Savings and Cooperative Banks, for small-scale solar PV and onshore wind in Germany (Geddes et al., 2018; Zhang, 2020).

The development of financial innovations is often necessary for financiers to support the new technology. Examples of symbiotic financial innovations include the creation and uptake of green bonds since 2006 and of ESG metrics in the last few decades to evaluate corporates, developed by a few incumbent financiers as a response to landscape pressure from a reputation crisis and shift in customer preferences (Monk and Perkins, 2020; Seyfang and Gilbert-Squires, 2019; Urban and Wójcik, 2019). Alternatively, financial innovations can be competitive and carried by financial entrants, leading, if successful, to a substitution of the financial regime in favour of financial entrants. Examples of the former include the creation of value-based banks such as GSL, Triodos, Ecology (Falcone et al., 2018; Seyfang and Gilbert-Squires, 2019; Urban and Wójcik, 2019; Yip and Bocken, 2018); of new banks replicating the example of local German banks in the UK to support small-scale investments in renewables (Abundance, Pure Leapfrog, Hampshire community bank) (Hall et al., 2016; Zhang, 2020) and of Fintech (Sánchez, 2022).

2.4. Extended theoretical framework

There are two main attempts to conceptualize finance within the MLP framework. For Geddes and Schmidt (2020), finance sits at the level of the regime, along with others such as the technology regime or the energy regime. Urban and Wójcik (2019) characterise finance as comprised of the 3 levels itself (financial landscape, financial regime, and financial niche where financial innovations develop) and the financial regime can undergo a transition. We build on the latter conceptualization of finance proposed by Urban and Wójcik (2019) (see Fig. 2).

Geddes and Schmidt (2020) and Urban and Wójcik (2019) framing failed to address certain limitations in the original depiction of finance within the MLP, which frames finance as a passive and neutral resource without agency. However, Naidoo (2020, 2020) demonstrated that finance is not neutral, but should be analysed through a behavioural lens. Moreover, neither of these frameworks adequately explain the diverse range of behaviours observed in empirical literature, nor do they account for the role of shipping financiers observed in our study. As a result, we propose an extended theoretical framework to explain the various roles financiers can take during a transition and the relationships between the industry and the financial MLP.

This diagram suggests a financial transition where new entrants carrying *competitive* niche financial innovations (level (3)) replace incumbent financiers (level (2)). In this proposed framework, as for industry transition, transitions might also unfold by the incumbent financiers (level (2)) adopting *symbiotic* financial niche innovations and therefore surviving into the new regime.

To do so, we build on the theoretical insights from the Adaptive Market Hypothesis (AMH) proposed by Hall et al. (2017) and Lo (2004) The AMH is grounded on path dependency and is helpful to understand the bounded rationality and decision-making of financiers. It assumes that the investment environment and financiers' behaviour change over time. In this framework, during times of



Fig. 2. Finance in the MLP framework. Taken from Urban and Wójcik (2019).

M. Fricaudet et al.

stability, financiers' investment decision process ("heuristics") are adapted to their environment and near-optimal, so that the Efficient Market Hypothesis, which is characterised by perfect information and rational expectations in the market (Lucas, 1978), broadly holds true. However, those heuristics become ill-suited when the environment changes, so that we observe "behavioural biases". Eventually, financiers adapt themselves to new realities in periods of turbulence or change, such as the Global Financial Crisis, which forces investor heuristics and behaviours to evolve gradually.

The concept of a heuristic is used both in the AMH and in the MLP literature, as they are both grounded in evolutionary literature. The MLP literature view heuristics as cognitive rules which are socially constructed and shared by the members of a socio-technical regime (Geddes and Schmidt, 2020; Geels and Raven, 2006; Geels and Schot, 2007; Safarzyńska et al., 2012). The AMH literature focuses at the individual level on the function of the heuristic, i.e. for an individual firm or person to make decisions under uncertainty (Lo, 2004). In this work, we build on both of these strands of literature and define heuristics as an adaptive set of tools, rules and beliefs which are socially constructed and shared among the members of a socio-technical regime who use them to make decisions under uncertainty. Those heuristics are sticky and evolving, i.e. they are fairly stable and cannot be changed overnight but they can evolve with time under landscape pressure.

However, we further propose 5 types of archetypal behaviours, depending on whether financiers are aware of the risks and opportunities of the transition, and on whether they can adapt to the new technology:

Inert: financiers anticipate stability of the existing industry socio-technical regime. For example, German banks and institutional investors were originally wary of financing renewables before KfW pro-actively gave positive signals towards those technologies (Geddes et al., 2018).

Creative self-destruction: financiers anticipate growth opportunities in the industry regime and steer industry incumbents to increase investments in the incumbent technology. This echoes the notion of Creative Self-Destruction proposed by Wright and Nyberg (2015) that the response to a problem arising from the continuous expansion of the industry incumbents is to further extend their reach. For example, E.ON and to a lesser extend RWE invested in the late 2000s' into fossil fuel electricity generation under the pressure of their shareholders (Ferguson-Cradler, 2022; Kungl, 2018; Kungl and Geels, 2018).

Loyal enabler: financiers anticipate the upcoming socio-technical transition and support the industry incumbents to invest in technology innovations (e.g. large support from the UK banks to the incumbent UK electricity utilities during the "dash for gas" to build gas-power plants, Bolton and Foxon 2015; Kern, 2012). The term Loyal Enabler is chosen because they support the transition by investing in the new technology but remain loyal to their existing customers. Several conditions are necessary for them to be able to do so:

- Either industry incumbents are already first movers, or financiers have the power to steer them in this direction; and
- They are able to finance the new technological innovation by adapting their heuristics.

Redirecting enabler: financiers anticipate the upcoming socio-technical transition and redirect their investments from industry incumbents towards industry niche entrants (e.g. support from the German Savings and Cooperative Banks to small-scale solar and wind in Germany, Geddes et al. (2018), Zhang (2020). This type of financier also supports the transition by investing in the new technology but contrary to Loyal Enablers, they redirect their capital from one industry player (incumbent) to another (niche) in its ideal-typical form. Financiers are able to do so because they can adapt their heuristics to those new entrants and/or the new entrants able to fit into their requirements, for example by agregating small-scale projects to fit into the transaction size desired by the financiers (those processes are described in Geddes and Schmidt (2020).

Winding down: financiers anticipate the upcoming socio-technical transition, but they are incapable of financing the new technology, so they reduce exposure altogether from the sector. That leaves the door open for other types of financiers to fill in the space.

This typology of behaviours can apply to both the regime and the niche levels of the MLP framework, i.e. to both incumbent financiers and financier new entrants, when they face financial and industry landscape pressure. One could assume that new entrant financiers would be more willing to finance industry niche entrants, i.e. play a Redirecting Enabler role in the typology (bottom right behaviour on Fig. 3). Similarly, one could assume that an incumbent financier would be more loyal to industry incumbents (e.g. existing shipowners), i.e. play an Inert, Creative Self-Destruction or Loyal Enabler role (two most left behaviours and top right on Fig. 3). However, there are examples in the literature of financiers supporting industry new entrants as discussed in the previous section, and there is no a priori reason why new entrant financiers would not also be resistant to changing their behaviour in an upcoming transition (Inert), so we assume here that both levels of the MLP can adopt all types of behaviour.

The above figure should be read from top left to bottom right. The black boxes represent the possible behaviours that financiers (either niche or regime) can adopt while an industry transition unfolds.

3. Methods

3.1. Case study selection

We conduct case studies of the incumbent financiers of shipping at the outset of two distinct socio-technical transitions: A transition towards low/zero-carbon shipping, to reduce the carbon impact of the ships themselves (supply-side risks").

Several socio-technical transitions of the land-based sectors, in particular the electricity generation and road transportation sectors, away from fossil fuels and towards renewables. This would translate into a decrease of transportation of fossil fuels such as coal and oil,



		Financial regime	
		Survival of incumbent financiers	Replacement of incumbent financiers
Beliefs on the industry socio- technical regime	Growth	Creative Self-Destruction	
	Stability	Inert	
	Transition by survival of industry incumbent	Loyal Enabler	Winding Down
	Transition by replacement of industry incumbents by niche entrants	Redirecting Enabler	

 \checkmark

possibly LNG as well; and potentially new shipping demand for the operation of offshore wind farms, the transport of CO_2 and bioenergy ("demand-side risks and opportunities").

The majority of ship finance historically comes from shipowners' balance sheet and debt provided by commercial banks, and more recently by Asian leasing agencies (Fig. 4). Debt issued by banks have tenors typically ranging from 5 to 12 years. Loan profiles are longer due to the economic lifetime of ships so that shipping loans include a balloon payment and are refinanced, often by the same bank. The Global Financial Crisis and the subsequent banking regulations changed the shipping financial regime, as they led European commercial banks to reduce their exposure from the sector because of the large scale of non-performing loans in shipping, in particular in Germany. In a sector characterised by a high volatility of revenues and asset values (see Fig. 5; see Alexandridis et al. 2018) for an in-depth review of the literature on the topic), remaining banks now focus on the largest and top-tier shipowners, which are perceived to be safer, resulting in intense competition in this segment.

The demand and supply-side shipping transitions that this paper investigates are just starting to unfold, and their outcomes are uncertain. It is not within the scope of this article to predict how they will unfold, but rather to look at the heuristics and ambitions of financiers at the beginning of them. The case studies therefore focus on whether they anticipate opportunities or specific risks to their portfolios from the demand-side and supply-side transition to low/zero-carbon shipping.

3.2. Methodological approach

We follow a qualitative research design, using the thematic analysis method proposed by Braun and Clarke (2006) and Clarke and Braun (2013). Since the research questions concern financiers' beliefs and perceptions of their environment, the principal source of data for this work are interviews. The understanding of heuristics and ambitions draws on data from 12 in-depth interviews with financiers, covering together around a quarter of the shipping debt, conducted between May and November 2022 (see Table 1). Eight interviews have been conducted with commercial banks active in shipping – including one State-backed bank with a mandate from the State – and 4 with asset managers, two of which are alternative lenders dedicated to the decarbonization of the shipping industry. It is worth noting that financiers 8 and 10 are effectively shipowners or ship managers who get some of their capital from asset managers. They provide a perspective on how asset managers and institutional investors view climate risks and make decisions. All interviewees but those two were mostly providing shipping debt to the industry, although some would also provide a range of products in addition to debt. Interviews were guided along a general interview guide but were semi-structured to allow the interviewer to alter questions depending on the interviewee's responses.

We further collected data from a range of secondary sources to give some context to the case study and triangulate the interview findings, which provides some observations of past investment decisions of shipping financiers and explores the public communication of the financiers concerning their role in climate mitigation. We collected data for the top 20 shipping banks according to Petropoulos (2021) and for all the financiers interviewed. News articles were collected by searching for the name of the bank and the keyword



Sources of shipping finance (\$ billion)

Fig. 4. Sources of shipping finance (from Tsianakidis, 2019).



Fig. 5. Prices on shipping markets, 1990-2022 (Clarksons Shipping Intelligence Network (SIN)).

"shipping", "marine", "maritime" or "Poseidon Principles" for all articles published after 2007. Only articles which were related to shipping low-carbon transitions and to a specific financial deal, when mentioning the ship(s) or type of ship financed, were selected. In addition, we collected ad hoc news articles to confirm and expand the understanding of specific points mentioned by the interviewees during the interviews when the context was not clear. We further collected public ESG/environmental/shipping reports and official communications related to shipping low-carbon transitions relating to those financiers, and the Poseidon Principles reports, using Google search. Finally, one financier agreed to provide non-public reports which they shared with their shareholders. We collected and analysed 113 company reports and official communications, 235 newspaper articles and 11 company reports shared with the companies' shareholders. Finally, quantitative data on past shipping investments were collected to provide context to the narrative.

Our data analysis uses the thematic analysis method proposed by Braun and Clarke (2006) and Clarke and Braun (2013) and is conducted in two steps. First, high-level themes were built using the proposed theoretical framework proposed in Section 2.4. Second, interviews transcripts and collected secondary data were mapped to those high-level themes by "coding" them using NVivo software¹ (top-down approach), but high-level themes were inductively adjusted to the internal logic of the data and detailed codes falling under each high-level theme were generated inductively (bottom-up approach). This second step is iterative, such the text is recoded as the detailed codes are iteratively developed. This method, also used in Kungl and Geels (2018), avoids artificial results which simply reproduce the theoretical categories. Both interview transcripts and secondary data were coded along the same high-level themes and detailed codes is presented in Table A1 in the appendix. For example, the codes "Customer and charterer demand for low or zero-carbon shipping", "ESG as legitimacy demand from society to financiers", "Shareholder demand for ESG" are grouped under the sub-theme "Society & customer demand for climate mitigation". Relevant text parts were coded along each subcode.

The data coded was primarily interpreted qualitatively, by looking at the context in which an assertion was made, the past history of the company of the interviewee, how certain or nuanced the interviewee is when making this assertion – for example several interviewees would state that "I believe that X is unlikely to happen, but if it does happen, then I believe Y", or whether it was made spontaneously by the interviewee or to answer a follow-up question of the interviewer. When relevant, this was complemented by some quantitative analysis on the number of interviewees who mentioned a statement coded under a theme, and by the number of times (words) an interviewee spent discussing this topic, which might reflect their interest or knowledge of the question. This quantitative approach is secondary to the purely qualitative approach.

The themes used to code the interview and the collected data are derived from the theoretical framework proposed in Section 2.4. A schematic representation of the themes is showed on Fig. 6, as follows:

1. A first set of themes cover the stickiness and evolution of the financiers' *heuristics*. We distinguish here between two types of heuristics, based on the coded data:

¹ NVivo is a software which allows the user to tag ("code") sections of texts with themes, and collate together all the text sections coded to a particular theme.



2. Behaviours

Fig. 6. Thematic analysis schematic representation.

- 1.1 *Beliefs*: Those cover financiers' beliefs of the drivers of the transitions (in blue on Fig. 6), such as the availability and risk of the incumbent and new technology and the strength of landscape pressures (policy, customer demand for example) behind the transition; and their beliefs on the role of incumbent shipowners (in green).
- 1.2 *Financial tools and instruments* (in blue on Fig. 6): those include the tools and rules of thumbs used by financiers (e.g. the credit risk methodology adopted by each financier; the rule of thumb used to judge the quality of a transaction; and the financial instruments available to financiers).
- 2. The roles of financiers in the transition to low/zero-carbon shipping are organised along the 5 ideal-typical *behaviours* proposed in Fig. 3 (central greyed area on Fig. 6).

The description of the themes and their underlying codes is detailed in Table A1. The following sections further explain the findings for both types of shipping transitions – demand-side and supply-side, and are organised along those three broad themes: *beliefs* on landscape pressures, financiers' *behaviour* in the transition and adaptation of the *financial tools & instruments*.

The numbering of themes is consistent with the one used in Table A1.

4. Results and discussion

4.1. Supply-side transition

4.1.1. Beliefs of the upcoming transition

All financiers interviewed acknowledged the need to decarbonize the shipping industry, which has been driven by landscape pressures for all regime actors to invest in energy efficient ships and low/zero-carbon ready ships. All interviewees mentioned that landscape pressures have already started to materialize through various channels and that they expect that trend to increase in the coming decade.

The type of supply-side transition driver perceived to be critical varied from financier to financier (see Fig. 7). Most financiers were particularly sensitive to changes in customer demand and increasing demand from charterers for cleaner ships. This customer and charterer pressure is linked to the adoption of the Sea Cargo Charter initiative in 2020, which establishes a framework for assessing and disclosing the climate alignment of ship chartering activities (Sea Cargo Charter, 2022). This initiative seems to have shifted beliefs of banks in particular, but also some alternative lenders, so that they now view the possibility of a two-tier market in favour of low-emission ships as credible in the future. Most financiers viewed regulation (at the regional level, in particular the EU, and in some cases IMO) concerning shipowners and financiers as an important driver. In particular, they anticipated further EU regulation on shipping emissions and increased capital requirements for financing carbon-intensive activities. Several anticipated increased IMO regulations in the long term, but many also mentioned that the institution was too slow or not ambitious enough. In particular, they anticipated further EU regulation on shipping emissions and increased capital requirements for financing carbon-intensive activities.



Fig. 7. Coded words and coding presence in codes related to beliefs of supply-side landscape pressures.

Concrete policies mentioned by the interviewees include the EU Emission Trading System (a cap-and-trade program that aims to limit greenhouse gas emissions implemented in 2005), the EU taxonomy (a classification system designed to define environmentally sustainable economic activities implemented in 2020), the CII/EEXI regulations (Carbon Intensity Indicator and Energy Efficiency Existing Ship Index), which are performance standards international maritime regulations that entered into force in November 2022 and are designed to reduce greenhouse gas emissions from existing ships. This suggests that recent regulations and announcements have shifted some financiers' beliefs about increased regulatory pressure. These financiers were optimistic overall of these potential policy changes and highlighted that such regulations are needed for shipping markets and financial markets to properly price in the climate impacts of shipping.

Many also highlighted societal pressure due to increased concerns about climate change, particularly since the Global Financial Crisis. Several interviewees highlighted the consequential demand from financiers' shareholders to improve their ESG policies, a demand which is also highlighted in their public communication, so that appearing to be a leader in climate change was cited as a competitive advantage by some of the interviewed financiers.

This apparent strong awareness of the coming transition to low/zero-carbon shipping needs to be nuanced by the scepticism expressed by all but two interviewees about the speed of the transition (see Fig. 7). Only one financier interviewed mentioned the decrease of shipping activity as a solution to reducing shipping emissions would potentially threaten the profitability of the shipping industry. Academic studies have shown that, if shipping activity is to continue growing as in the previous decade and shipping to remain in line with a 1.5° pathway, ships' carbon intensity needs to be reduced by at least 50 % by 2030 (Science Based Targets, 2022; Traut et al., 2018) and low/zero-carbon alternative fuels to be rapidly scaled up in the 2030s (Lloyd's Register and UMAS, 2019; Osterkamp et al., 2021). This ambition was deemed unrealistic by many financiers "There was a discussion in Poseidon [Principles] about this new potential new trajectory based on top-down (...) carbon budgeting, implying that (...) we need to take 50% of the reduction in the carbon intensity (...) before 2030. That's totally unrealistic." (Interviewee 9). When directly asked, many would acknowledge the existence of a technology risk but would often consider it as less important as it was felt that the technology for low/zero-carbon ships was uncertain and immature while it would take a long time to renew a whole fleet "It's in the pace of the creation of future fuels that may well determine whether a particular asset is viable forever till the end of its useful life or you know its life it's foreshortened and in the sense that it comes too expensive to retrofit and stuff. So these are fortunately not decisions that I will be making. (...) But I think the risk of stranded assets in shipping is quite low. 'cause I think if ships don't get finance they don't get built, so no one wants to order them. But of course they could get stranded if the pace of technological change turns out to be much faster" (interview 1). Furthermore, several interviewees expressed technology lock-in for LNG, since LNG as a marine fuel has been shown to have limited GHG benefits (Balcombe et al., 2022; Pavlenko et al., 2020) and its drop-in low/zero-carbon fuels such as bio-LNG and e-LNG are not expected to be available at scale nor cost-competitive compared to other low/zero-carbon fuels (DNV GL, 2020; IRENA, 2019, 2021; Mærsk Mc-Kinney Møller Center for Zero Carbon Shipping, 2021; Smith et al., 2019). Interviewed banks, all of which have already invested in LNG-fuelled ships (including LNG carriers) were however putting faith in drop-in fuels such as bio-methane or e-LNG. Arguments mentioned by interviewees for continuing to back this technology included the trust in the shipowners' judgement, belief in the availability of drop-in zero-emission fuels, lower SOx, NOx and CO₂ emissions compared to LSHFO, and confidence that engine improvement will solve the issue of methane slip.

Only a few financiers on the other hand expressed concerns that the assets they finance – both conventional and low/zero-carbon – would face early obsolescence due to the uptake of new propulsion technologies: "Today we're approached a lot of projects to finance (...) LNG-fuelled or methanol, that we didn't see before. Tomorrow we'll be approached on ships to finance with hydrogen or ammonia



Fig. 8. Words coded and coding presence in the codes related to beliefs on shipowners.

M. Fricaudet et al.

and that means that the current fleet will lose value faster than we could expect. And then these new ships, well, we don't really know what their value will be (...) in 10, 15, 20 years because they will probably be out of date too. So it's a real period of uncertainty" (Interview 7).

Interviewees are counted if the interviewee has expressed a statement in line with the code.

Most of the financiers believed that the top-tier incumbent shipowners, which constitute the bulk of their clients, are aware of the need to decarbonize and are first movers in the transitions (see Fig. 8). This finding is valid whether measured by count of words of by count of interviewed who expressed an opinion on the shipowners in line with one of the 5 types of behaviours proposed in the extended theoretical framework. The Poseidon Principles reports are full of such statements: "We thank our shipowner clients who have been very cooperative for this first "harvesting" data campaign concluded in a very tight schedule. This is the clear demonstration of their commitment to a greener shipping and their willingness to be actors of the energy transition for shipping" (CIC, Poseidon Principles 2020, 26). This view is largely supported by the interview data, although many interviewees would nuance this statement by stating that some shipowners are reducing their investments in any ship due to uncertainties about their future viability ("regime shipowners are Winding Down exposure" on Fig. 8) or that some shipowners are ready to move earlier than others ("Regime shipowners = Inert" in Fig. 8). In particular, it was felt that public shipowners and shipowners operating on shipping segments closer to customers (ferries, containers, offshore) were more advanced, which is in line with the findings of Stalmokaite and Yliskyla-Peuralahti (2019). With regards to the beliefs that their clients are first movers and that the decarbonization will be gradual, many financiers believe that in case of the materialization of climate risks, their clients' ships will not be stranded because they are the most modern on the market and because the loan period is relatively short (around 7 years) so that they will have the time to adapt to the new regime, while other ships will be stranded first. It is therefore noteworthy that no incumbent financier considered that their clients were making value-destructive decisions ("regime shipowners = Creative Self-Destruction" on Fig. 8) or that new entrants would be better placed to take on the opportunities of the low-carbon transition ("Niche shipowners = first movers", on Fig. 8).

For most interviewees, investment in low/zero-carbon ships requires access to a varied and secure source of capital and the involvement of a large range of stakeholders to share the financial risk and additional costs – for example customers, charterers, fuel providers, public authorities and ports. Most banks interviewed believed that only large shipowners – which are also generally their target clients - will have the financial, relational and experience capabilities to do so. Some expected – and welcome – a consolidation of the market around the few top-tier shipowners they are financing. In particular, financiers – whether commercial banks or asset managers shipowners – were keen to pass some of the risks associated with new technologies to the charterers. This identified need has not translated so far into a financial innovation from banks, but two asset managers have highlighted that their business model was based on long time-charters and one of them is trying to add a clause in the contract which stipulates that the charterer would have to buy the ship back at the end of the charter period, which essentially displaces the residual risk from the shipowner/financier onto the charterer.

The counts corresponding to "niche shipowners = first movers" come from interviewees 8 and 10 who referred to themselves when saying that niche entrant shipowners were first movers.

4.1.2. Role in the transition

Given the previously mentioned beliefs about supply-side risk, interviewees expressed an ambition to play a Loyal Enabler role, i.e. supporting incumbent shipowners in the transition. Shipping financiers have publicly communicated this ambition, in particular since 2019, in their annual reports and public communication but also through their involvement in various initiatives such as the Sustainable Shipping Initiative, Ship Recycling initiative, the Global Maritim Forum or the Poseidon Principles.

The Loyal Enabler behaviour seems to derive directly from the historical links between the financial regime and the industry regime through relationships with their clients which are often personal, in which they have invested in incumbent technologies, and common beliefs about the future availability of fuels. All banks interviewed considered ship finance as corporate finance rather than asset finance, i.e. finance to the shipowner as opposed to an individual ship asset. As a result, the importance of the ship as collateral often comes as a second or third priority. The focus on client was driven by credit worthiness of the borrower and in some cases to also ensure that the company fits into the financier's business strategy. A particularity of ship finance is that banks attach a large importance to the reputation, performance in past shipping cycles and the relationship with their clients, at least as much as to the current financial health of the shipowner company. This behaviour is corroborated by previous findings in the literature (Gavalas and Syriopoulos, 2015; Lee and Pak, 2018), and interviews support that this is still one of the main drivers of a financing decision. Several banks stated that those links would allow them to positively influence the evolution of the industry, through engaging with their clients and through making capital available for cleaner ships. On the other hand, many banks seemed cautious to support niche entrant shipowners, but the interview with interviewee 10, a shipowner operator who has entered the ship-owning space a few years ago and who differs from other incumbent shipowners by their mandate to only invest in ships exceeding the climate ambitions of the Paris Agreement, contradicts this statement, as they have been able to access finance from traditional shipping banks. However, the interviewee highlighted that the fact that their company's management entrant had decades of experience in the shipping industry and were personally known to the bank employees helped build the relationship.

The belief of incumbent shipowners as first movers, and the fact that all banks interviewed viewed their business as relationshipdriven means that banks largely trusted their clients' judgement on the selection of ships in their fleet and their ability to transition to a low/zero carbon fleet. Although some have highlighted some reservations in the interviews, this trust remained strong: "So your question as to what we look at for new vessels [*in terms of carbon efficiency*], I would have said we listened to our client (...). But now we know that if we agree this science-based target, we can't make that assumption any longer. If it's an LNG-fuelled vessel, we know actually that it will make our results worse. That won't necessarily stop us financing them there because we're taking the

M. Fricaudet et al.

decision to support our clients who are making the assumption that they will be able to run these LNG fuelled vessels on bio or synthetic LNG in years to come, which is a big... It's a mighty assumption, but it's one which a lot of these the sector is taking" (interviewee 2).

This approach of shipping finance as relationship-driven corporate finance was largely shared by the banks, but not necessarily by the other types of financiers interviewed, due to the difference in business model. Institutional investors were found to attach a large importance to transparency, corporate financial and increasingly ESG metrics, but not to relationships. The financing behaviour of these interviewees can be characterised as a Redirecting Enabler, i.e. anticipating the upcoming socio-technical transition and redirecting their investments from industry incumbents towards industry niche entrants. On the other hand, the alternative lender 3, despite a much more ambitious climate agenda, had a business model similar to traditional banks based on secured debt lending and with a large attention to the reputation of the shipowner, although a major difference is that they put the ship characteristics, in particular climate-related ones, before the company.

For some financiers, their ambition to play a Loyal Enabler role has started to translate into investment decisions in low carbon shipping assets. Two financiers reported having financed pilot projects of low/zero-carbon ships to improve their own knowledge of niche technologies and provide proof of concepts to the broader industry. It is worth noting that for both of those projects, the role of the State mandate and public finance was prominent. Several other banks mentioned they were wary of providing finance for unproven technology and would rather step in once the proof of concept has been established. The impact of this shift in investment decisions on the shipping portfolio is however marginal for all financiers interviewed but the two alternative financiers at the time of writing. Furthermore, many of the interviewees have provided finance to LNG-fuelled ships. The case of LNG-fuelled ships is insightful because it illustrates how financiers – in particular shipping banks – can play a Loyal Enabler role to support the uptake of this new technology by their incumbent clients; but also good case demonstrating inertia and technology lock-in. Interviewed banks which have already invested in LNG-fuelled ships were reluctant to move away from those investments despite the risks (Balcombe et al., 2022; Pavlenko et al., 2020).

4.1.3. Adaptation of financial tools and instruments to supply-side climate risk

Traditionally, most of the commercial banks' and financier 3's activity is debt secured by the ship asset with longer profile than tenor, i.e. the ship is depreciated along a longer period than the loan duration so that there is a balloon payment at the end of the loan period. Credit rating and investment decisions are mostly based on backward-looking financial (non-climate related) data such as profitability or leverage; where corporate metrics are prevalent while the asset characteristics remain secondary. Most financiers' risk management approach is to finance newbuildings over refinancings, or ships up to 10 years old. This way of working was found to persist in the interviews, but banks have started to collect environment-related data over the last few years under the landscape pressure to decarbonise.

First, interviewees highlighted that while fuel efficiency was originally not taken into account in investment decisions by financiers – validated by previous literature which found that the most relevant metrics for fuel efficiency were not taken into account in 2016 (Mitchell and Rehmatulla, 2016), they have now started to collect data on the environmental characteristics of the ship assets they finance. The main data collected is the Annual Efficiency Ratio (AER, a proxy of energy efficiency which assumes ships are fully laden on each voyage) of second-hand ships because it is the metric used in the Poseidon Principles, but some financiers mentioned looking at the design fuel efficiency, the fuel used, the age of the ship, the capacity to retrofit to alternative fuels and for a few the risk that a ship becomes stranded using qualitative assessment or basic desktop analysis. For example, one interviewee's approach was to have a distribution of ages, including older ships as these would leave the fleet in the short-term. They were wary of newer ships because technological and regulatory uncertainty might cause these ships to be outcompeted in the market with ships which will be more technologically advanced (e.g. in terms of fuel and machinery onboard) and compliant with regulations.

Second, the perceived landscape pressure to decarbonise the shipping industry also translated into an increasing attention to the decarbonization strategy and ESG characteristics of the shipowners. As a result, banks have continuously been putting efforts over the last few years to develop tools to measure the ESG performance of the corporates. Those tools might incorporate, among other variables, the carbon intensity of the total fleet, but would not reflect the particular risk attached to an asset: "It's not that we won't finance assets above the pathway, but we want to make sure that our owners have a strategy to decarbonize their fleet and particularly the ships that we're financing" (Interviewee 4).

How this newly collected data influences financier's actual behaviour and investment decisions is ambiguous. First, this datadriven approach is combined with the belief that the top owners (who they provide finance to) are most resilient to the energy transition and own the most fuel efficient ships. Second, for all interviewees but the two alternative financiers, those ship-level characteristics did not seem to be formally included in the financial assessments and credit rating used to inform the pricing and financing decisions. In particular, they did not incorporate carbon prices or future CAPEX investments for low/zero-carbon fuel switching retrofits. As a result, the way the environmental performance of the ship assets feeds into the credit rating and the pricing was not formally and explicitly defined, even when this environmental performance is measured and quantified in a scenario analysis. As a result, debt pricing did not reflect the climate risk of the ship, but rather the competition between banks to provide debt finance to the few top-tier shipowners they are after.

Although the ships' environmental performance did not influence the conventional financial instrument's pricing, many financiers highlighted that the perceived greenness of the ship would influence the engagement with the shipowner, and in some case the decision to provide finance to a ship, in particular after signing the Poseidon Principles. For one bank and for the alternative financiers, climate risks were incorporated into guidelines on the carbon intensity of the ship above which they would not invest, but that position seemed to be marginal within the sample banks, which seemed to have a more qualitative and flexible approach and they would not refuse to finance a ship to their existing clients on the basis of its carbon intensity. As mentioned, a central characteristic of ship finance

is the importance of the relationship with the shipowner corporate, this characteristic has been simply reconfigured to accommodate growing landscape pressure, and has not been overruled by the need to finance low/zero-carbon ship assets.

To formally incorporate the shipowners' and/or the ships' environmental performance into a financial deal, many financiers interviewed have started to utilise various types of green finance, where the margin fluctuates depending on the operational efficiency of the ship financed or other environmental characteristics of the borrower or the ship (e.g. sustainability-linked loans or green bonds). However, green finance in shipping suffers from the same criticisms as other industries. First, they only represent a small part of the bank activity: environmentally-linked bonds represent 5 % of released bond amount between 2019 and 2022 reported in Clarksons Shipping Intelligence Network (SIN), while \$10.5bn of shipping environmentally-linked loans have been issued so far, i.e. roughly 4 % of the bank shipping debt portfolio². Second, those issuances are always released in agreement with the shipowners and several interviewees admitted that the effects on the margins are fairly small so far, making them only marginally more attractive from a commercial point of view. Third, there is no agreed definition of what a green shipping transaction is. In particular, less than 5 % of the green issuances reported in Clarksons SIN have been approved by the Climate Bond Initiative, the International Capital Market Association (ICMA) or the Loan Market Association (LMA), which casts doubts on how actually environmentally-friendly the remaining issuances truly are. Such transactions have been accused of green washing by some NGOs (ShareAction, 2019; Wiese Bockmann, 2022) but also by one of the shipping asset managers interviewed. Similar to other sectors, green finance in the shipping industry seems to constitute an incremental financial innovation as a response to financiers' landscape pressure to increase their legitimacy by visibly demonstrating an environmental contribution (Monk and Perkins, 2020). Green finance transactions are publicized in the press and companies' public reports so that the lender and the shipowner benefit from positive publicity. Their uptake since 2019 in new debt issued is shown in Fig. 9. Considering that there were practically no green financial instruments for ship finance before 2019, this shows that there is some interest from both lenders and issuers towards this kind of instruments. All interviewed banks had already issued at least one green issuance, which suggests that the uptake covers a large range of banks rather than a few first movers (Figs. 10 and 11).

Only bonds and loans reported in Clarksons SIN whose company sector is classified as shipowner or integrated cargo/shipping group are included. Green issuances cover a large range of bond and loan types which are often not approved by the Climate Bond Initiative (CBI), the International Capital Market Association (ICMA) or the Loan Market Association (LMA), so that the definition of what is "green" varies from issuance to issuance. New shipping loans amount is proxied as the new bank debt issued in 2019 (BRS Group, 2019) and is assumed to be constant in the following years in the absence of better source of information.

4.2. Demand-side transition

4.2.1. Beliefs of the upcoming transition

Overall, financiers did not expect any stranded assets of the fossil carriers they finance in the short-term, as they saw no driver for an immediate and strong decrease in fossil use in land-based sectors and they viewed fossil fuels as necessary to the energy security of countries. Those expectations did not significantly differ by the type of financier: even the alternative financiers who were refusing to finance any coal or oil carriers out of principle grounds did not believe the fall in oil consumption would be dramatic in the short-term. Interviewees believed that coal shipping demand would decrease eventually but that it would not have a large impact on the bulk shipping demand because bulk carriers transport multiple cargoes. They expected the demand for oil shipping to remain strong in the coming decade and maybe decline afterwards, but some acknowledged that there is a risk of stranded assets in the tanker market if its demand declined earlier than this. On the other hand, many financiers expected LNG transportation to remain strong or even increase in the coming decade, driven by its perceived transitional role for decarbonization of the land-based sectors, and geo-political factors such as Russia-Ukraine war, which has led to an increased demand for LNG transportation.

Those beliefs were at odds with the public commitments of most of the major shipping banks to align their finance in land-based sectors to the Paris Agreement, which requires a rapid phase-out of fossil fuels in the coming decade and would likely drive the demand for fossil fuel shipping down. In particular, most of the shipping banks and most of the interviewed banks are signatories of the Net-Zero Banking Alliance, through which they have committed to align their portfolio to a pathway consistent with a maximum 1.5° pathway. This suggests that shipping financiers might not have fully grasped the implications of being aligned with a 1.5° pathway.

This belief of the non-materiality demand-side risks is not communicated publicly, and this issue is largely ignored in the public communications of financiers. A non-public market report sent by one financier to their shareholders, and to which the authors had access, mentioned that oil & gas transportation demand would remain strong. This suggests that opposing views were shared privately and publicly, although it is not possible with the current data to generalize to all financiers.

On the other hand, a few financiers mentioned that the decarbonization of land-based sectors were bringing significant business opportunities. Demand-side opportunities mentioned by the interviewees includes the transport of CO_2 , hydrogen, ammonia, bioenergy, and maintenance of offshore wind. Alternative financiers were particularly enthusiastic about such opportunities.

4.2.2. Role in the transition

As a result of the belief that demand-side risk is not material – with some exceptions, most financiers were continuing to invest in fossil fuel carriers, in particular oil and LNG carriers, and were not planning to stop doing so in the short-term: "I have no objection to

² Calculated based on the top 62 shipping banks portfolio reported in Petropoulos (2021).



Fig. 10. Words coded in the codes related to beliefs of demand-side risks.

financing a tanker though I would be reluctant to finance a tanker that wasn't dual-fuelled or retrofittable because they're still going to be moving oil." (Interviewee 1). This corresponds to an Inert behaviour. In this context, the only financiers of the sample who have fully refrained from financing oil tankers were the alternative financiers. This finding should however be nuanced by small signs of Winding Down of certain types of oil tankers by a few financiers, which were considered to be particularly at risk. In particular, 3 banks interviewed have mentioned being reluctant or refusing altogether to finance Very Large Crude Carriers (VLCC) while another one mentioned requesting shorter repayment profiles for financing oil tankers. Similarly, as per supply-side risk, some interviewees expressed the belief that their portfolios were not at risk because, as the ships they have invested in are the most efficient on the market, they would be stranded last, so that when oil shipping demand starts declining, they would have the time to divest from oil carriers without incurring losses. Moreover, most financiers are wary of financing bulk carriers dedicated to coal transport - but as most bulk carriers can carry other commodities as well, this rule was not restrictive in practice.

Because some financiers were also particularly optimistic on the future opportunities of LNG shipping, many have adopted an aggressive investment strategy for LNG carriers. This behaviour can be considered as "Loyal Enabler" if the demand for LNG shipping demand materializes, or "Creative Self-Destruction" if it remains steady and declines. It is not possible to conclude between the two

	Heuristics		Behaviours during
	Beliefs of the	Financial tools and	transition
	transition	instruments	
Supply- side risks	 Most believe transition to low/zero- carbon shipping is happening. But it will not be fast enough to be aligned with a 1.5-degree pathway. Given above and tenor, financed ships are thought not to be at risk of being stranded 	 Currently measurement lies on corporate ESG score instead of asset/ships. Measurement and reporting of ships' carbon intensity is based on proxies. Ambiguous use of green finance (sustainability- linked, Poseidon- Principles-linked, Green Bonds Principles) Willingness to transfer of risk onto the charterer 	 Historically most can be classed as lnert (anticipate stability) Many have ambition for becoming Loyal Enabler (anticipate and support the transition) This varies by type of financier: traditional banks, due to the importance of the relationships with their existing clients, are more inclined to be Loyal Enabler. It is not clear yet whether alternative financiers and institutional investors will play a Loyal Enabler or enabling role. Some face risk of Winding Down (incapable of operating in transition)
Deman d-side risks	 Demand-side risks are not perceived to be material. Coal shipping demand will decrease slowly. Oil tanker demand is perceived to be strong for the coming decade and will decline thereafter. LNG shipping demand perceived to remain strong or even increase 	Exclusion rules on ship types by a minority of financiers	 Largely Inert for coal and oil transportation Loyal Enabler for LNG transportation

Fig. 11. Summary of results.

types of behaviour at the time of writing, as there are large uncertainties on the future policies' climate ambitions, for example inclusion of well-to-wake emissions in international regulations, and even if the land-based sectors aligns with a 1.5° pathway, there is no consensus on the consumption of LNG aligned with a 1.5° pathway, so that the future demand for LNG shipping is uncertain (See for example contradicting scenarios outputs from IEA 2021, IPCC 2022, Sharmina et al. 2017, Walsh, 2019).

4.2.3. Adaptation of financial tools and instruments to demand-side climate risk

Because of the beliefs that demand-side risks are not material, heuristics related to demand-side risks have been very stable. A few financiers have invested some resources in exploring demand-side risks and opportunities, in particular to understand which specific ships are most at risk of being stranded. "I think it's all only if you managed to push the curve down to 1.5–1.4°, if you really manage that, you will not (...) continue building oil tankers. That's our a very brief desktop analysis" (interviewee 9). The resources invested seemed to be limited and the majority of banks did not mention performing this exercise. Furthermore, only one financier has suggested it would look at the potential alternative use of its ship if the demand for its cargo carried were to reduce. A minority of financiers have made the binary choice of excluding oil tankers, and more coal-dedicated bulk carriers from financing.

4.3. Summary of results

Traditional shipping lenders, who are part of the shipping financial *regime*, have been until recently, inert to climate risks and opportunities in the shipping industry. However, our results show that the accumulation of *landscape* pressure (existing and upcoming

shipping and financial regulation, customer and shareholders pressure due to an increased climate change awareness) have led them to change their *heuristics* partially and imperfectly. In particular, they are confident that a supply-side shipping transition is underway, a trend which developed in the last few years. However, they are more conservative about the speed of the transition than what would be required to attain a 1.5° pathway and avoid the dangerous consequences of climate change. They are even more sceptical that the materiality of the risk that demand-side risk materializes, i.e. that the demand for shipping fossil cargo decreases rapidly.

As a consequence, they are not particularly worried about stranded asset risk on the fleet they have financed and continue financing fossil-fuelled ships and fossil carriers. However, they have started to change some of their financial instruments to reflect their growing environmental awareness, firstly by looking at the environmental performance of the borrower and secondly, by starting to take the environmental characteristics of the ship into consideration. This supports the theoretical framework of the AMH that during periods of change and landscape pressure, financiers' *heuristics* do evolve over time but are sticky and slow to adapt.

One heuristic which has been notably stable is the importance of the relationship with and the trust in the judgement of their existing clients, i.e. the incumbent shipowners who are part of the industry *regime*. As a consequence, the financial and the industry *regimes* appear strongly supportive of each other and traditional lenders aspire to support them in transitioning by adopting innovations (LNG or alternative fuelled, LNG carriers), i.e. to play a *Loyal Enabler* role. At the time of the interviews, there were few signs that they would want to support innovations which would require supporting niche shipowners (*Redirecting Enabler*) or that they felt they should retreat from the sector (*Winding Down*). This might give an advantage to regime incumbents to undertake the transition, while making the success of innovations more difficult, and suggests that innovative shipping niche players might need to seek alternative sources of finance.

5. Conclusions

Shipping financiers, like other stakeholders in the shipping industry, are facing a growing pressure to reduce their exposure to high polluting shipping assets due to the ratcheting up of climate policy in the EU and at IMO and from a wider societal pressure to decarbonise the industry. In reaction to this pressure, voluntary private stakeholder initiatives like the Poseidon Principles were created to measure and disclose the climate alignment of their portfolios. This has been a catalyst for the industry to become more aware about climate risks and incorporate new factors into financing decisions and risk management practices.

In this paper, a new and extended theoretical framework was created to understand the role of financiers during sustainability transitions. This builds upon the Multi-level Perspective to describe an energy transition and Adaptive Market Hypothesis to describe how financiers adapt their expectations following a breakdown in heuristics. The framework uses questions to deduce the type of financier, characterising financiers into five archetypes to take into account the wide range of financing behaviours observed in the empirical literature of other sectors and their adaptive nature which is not taken into account in extended MLP frameworks. Second, this paper provides novel empirical insights on the beliefs and ambitions of shipping financiers at the outset of several sustainability transitions in shipping regarding their future role and the risks to their assets using qualitative thematic analysis and semi-structured interviews with ten shipping financiers and two shipowners. The findings from these interviews were used to characterise shipping financiers according to the adapted MLP financier framework.

The results show that the majority of interviewees expressed an ambition to play a Loyal Enabler role, i.e. supporting incumbent shipowners in the transition, while a second type of financier – the Redirecting Enabler - emerged from interviews with niche shipowners receiving funding from institutional investors. The finding that the majority of financiers interviewed (and all commercial banks) can be categorised as *Loyal Enablers*, i.e. they are supportive of incumbent shipowners which is their customer base, makes it more likely that the energy transition will be largely driven by incumbent shipowners. Most financiers also place the responsibility of the energy transition with the shipowner, which could lead to further technology lock-in (exemplified by some banks' support of their clients' LNG-propelled ship investments) if financiers do not investigate the associated climate risks. This also implies that niche shipowners will not represent a large proportion of the fleet but as first movers, they can put pressure on incumbent shipowners and derisk technologies that conservative shipowners are unwilling to be exposed to. Financiers' support of fossil fuel cargo e.g. oil tankers, reflects their scepticism of overall demand-side risks materializing, i.e. that the demand for shipping fossil cargo decreases rapidly. This could delay the transition and cause a more difficult transition later on when deeper emissions cuts are necessary due to the cumulative impact of GHG emissions on climate change.

Financiers are adapting to the transition through more stringent requirements, e.g. expecting clients to have a credible transition plan, pay loans off more quickly, own modern vessels, and for low/zero carbon ships to have a contract in place in an attempt to transfer the transition risk onto the owner and charterer. This increased burden implies a trend of consolidation of the market towards owners at the top tier who can invest in transition plans and have a strong cash flow to invest in R&D for greener fuels. There is also a need for more finance from Redirecting Enablers to accelerate the transition which is likely to come from alternative funding sources like institutional investors.

Financiers and shipowners should assess the climate risk of their portfolio (fleet) and ensure that the ships in their portfolio (fleet) are climate resilient and financiers' clients have a credible transition plan (e.g. through investing in future fuel ready ships that have the optionality of running on low/zero-carbon fuel and avoiding marginal investments like LNG that have high cost fuel substitutes). However, there are limitations on how much the industry can drive the transition without more stringent climate policy. Among European banks, there is a sense that regulation is needed for banks to properly measure climate risks, such as the EU taxonomy and differentiated capital requirements for green and brown assets. Several European banks highlighted that regulation of financial reporting and capital requirements were needed for banks to fully play their Loyal Enabler role, as the market competition between banks was preventing them from fully incentivizing greener technologies. This highlights the potential, but also the limits, of voluntary

M. Fricaudet et al.

initiatives from financiers to promote green investments. Furthermore, pilot projects undertaken by commercial banks were in collaboration with the State, underlining the importance of the role of the State or international finance institutions in sharing financing risk through loan guarantees or financing support.

This study has a few limitations. Although the banks interviewed represent around a quartier of debt financing, it covers largely the views of European and Japanese banks who could be said to be leading on integrating climate considerations into their lending decisions. It does not capture a large and growing area of financing in China. Second, the sample largely represents debt financing from banks rather than alternative financing from institutional investors. The findings from these alternative lenders could be investigated further using a larger sample given their key role in financing the emergence phase of the transition. Future research could therefore attempt to understand the beliefs of financiers diversified across geography and the type of lending (e.g. financiers to smaller shipowners and alternative lending).

Declaration of Competing Interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Data availability

Data will be made available on request.

Acknowledgments

This project has received funding from Baltic Exchange (PhD funding).

Appendix

Interview guide Investments decisions – descriptive

- 1. What types of financial products do you provide to shipowners?
- 2. How long is the tenor and the profile typically?
- 3. What type of ships and clients do you finance?
- 4. If you had to give 3 main factors you consider when deciding whether or not you will provide finance for a ship, which ones would they be?
- 5. If you had to give 3 main factors which influence the interest rate you give, what would they be?
- 6. Why signing the Poseidon Principles?

Evolution of the industry over the last decade

- 1. Could you tell me the story of the first ship investment that you made in your carrier?
- 2. Could you tell me the story of the last ship investment that you made?
- 3. Have you observed any evolution in the way your company views and mitigates for climate risks since you joined?

Expectations concerning future stranded assets

- 1. How do you feel the demand for cargo shipping such as oil, coal and natural gas will evolve in the coming [*insert tenor] years?
- 2. How likely do you feel it is that it will impact the value of the fleet you finance?
- 3. How do you feel the pressures to limit carbon emissions from shipping will evolve in the coming [*insert tenor] years?
- 4. How likely do you feel it is that the ships you finance lose their value because of efforts to limit carbon emissions from shipping?5. How do you mitigate for those risks (if at all)?
- 5. How do you miligate for those risks (if at all)?
- 6. Under which conditions would you finance alternative-fuelled ships?

Detailed description of themes and codes

Table A1

Description of the themes and code.

Theme/code

Description

Table A1 (continued)

Theme/code	Description
1. Heuristics	Evidence of sticky or evolving heuristics. Heuristics include, as per the theoretical framework, believes and financial tools & instruments.
1.1. beliefs	A per the theoretical framework, those cover financiers' beliefs of the drivers of the transitions, such as the availability and risk of the incumbent and new technology and the strength of landscape pressures (policy, customer demand for example) behind the transition.
1.1.1. beliefs on landscape pressures	Financiers' perceptions of the existing and upcoming exogenous pressures to shipping.
Cyclicality of shipping markets	Even without considering climate risks, the cyclicality of the shipping revenues means that financiers are exposed to the risks that shipowners go bankrupt of non-performing loans and devaluation of underlying ship assets.
Financiers' awareness of climate risks has increased Target to reduce shipping emissions	There has been a general increase in financiers' awareness of climate risks over the last few years. The financier has published a target for reducing emissions of its shipping portfolio. Note that the targets are not homogenous and not all aligned with a 1.5° pathway.
Further regulation and public support necessary for the transition	The private sector cannot self-regulate climate risks and need some kind of regulatory or financial support from the public institutions.
IMO failure to act fast enough	The IMO has been slow to regulate shipping emissions and more efforts are required.
Role of public finance	Instances where public finance has been an Redirecting Enabler of the transition.
Increasing regulatory pressure on emissions	The financier perceives increasing regulatory pressures on shipping emissions.
Society & customer demand for climate mitigation	Growing demand from society and stakeholders for climate mitigation of shipping emissions and/or land-based sectors decarbonization.
Customer and charterer demand for low or zero-carbon shipping	Customer and charterers increased demand for low/zero-carbon shipping is a driven of profitability of low/zero-carbon ships.
ESG as legitimacy demand from society to financiers	Financiers perceive that there is social demand towards financiers for ESG/responsible investment.
Shareholder demand for ESG	Equity investors have started to require ESG/climate consideration from financiers and shipowners.
recimology fisk	rmanciers are worried that snips might become obsolescent earlier than they have historically.
only finance modern ships	the interviewee's company only infances the most modern, fuel-enciency of carbon-enciency sings so they do not believe they will be at risk of being stranded
Demand-side risk is not material	Because there is no expected large decrease of fossil fuel shipping demand, fossil fuel carriers are not at
	risk of being stranded.
Demand-side risk is material	The financier believes that fossil fuel carriers are at risk of being stranded.
Financiers' ambition to reduce the emissions of their	The financier has committed or expressed the ambition to align their land-based sectors portfolio (e.g.
land-based sectors portfolio Shipping opportunities from the decarbonization of land-based sectors	energy, manufacturing, etc) to a decarbonization trajectory. The financier ambitions to support shipping demand linked to renewables, e.g. offshore wind.
1.1.2. beliefs on shipowners	This theme regroups all statements characterising the behaviour of shipowners.
Niche entrant shipowners = first movers	Financiers perceive that emerging niche actors are taking up the opportunities arising from the transition to low/zero-carbon shipping and the decarbonization of land-based sectors.
$Regime \ shipowners = Creative \ Self-Destruction$	Financier perceive that shipowners anticipate a growth in shipping and no significant decarbonization, hence increasing investments into the existing technologies.
Regime shipowners = first movers	Financiers perceive that incumbent shipowners are taking up the opportunities arising from the transition to low/zero-carbon shipping and the decarbonization of land-based sectors.
Low or zero-carbon shipping will lead to a consolidation of the market	The shipowners believe that the transition to low/zero-carbon shipping will favour the largest historical shipowners, leading to difficulties from the smaller shipowners and a consolidation of the market.
Regime shipowners = Inert	Financiers perceive that some or all shipowners are unaware of the risks and opportunities arising from the transition to low/zero-carbon shipping and of the decarbonization of land-based sectors.
Regime shipowners are Winding Down exposure	Financiers perceive that shipowners are reluctant to invest in new ships today because of the uncertainty of the transition.
1.2. Adaptative of financial tools instruments	As per the theoretical framework, those include the tools and rules of thumbs used by financiers to make decisions. Those include for example the credit risk methodology adapted by each financier; the rule of thumbs used to judge the quality of a transaction; and the financial tools and instruments available to
1.2.1. Sticky heuristics - backward-looking risk management	financiers. Backward-looking heuristics and tools which have been traditionally used by shipping financiers to assess the quality and the risk of a transaction continue being used in the transition.
Quantitative backward-looking risk assessment	Standardized quantitative credit risk assessment, based on historical backward-looking data.
Standardised financing instrument	The type of financial product is standardized within and across financiers
1.2.2. Type of business = corporate financing	This set of codes look at how financiers view shipping finance as corporate finance, i.e. finance to a shipowner, rather than asset finance. Concretely, when assessing whether or not the finance will finance a ship, it will look at the characteristics of the shipowners before looking at the characteristics of the
Analysis of the exect is seened in the	ship asset.
Analysis of the asset is secondary	The mininger either does not look at the asset, or looks at it after having looked at the corporate.
Prevalence of corporate non climate-related metrics	Financies for only provide sing secured devision and range of products to a few chefts. Financiers primarily use financial (non-climate-related) corporate metrics which do not include climate risks on the shipowner to judge of the quality of a deal
Shipowners' reputation is an important criterion of decision	Qualitative criteria related to the shipowners' reputation are used in financiers' assessment process.
Shipping finance is relationship-driven	The financier has long-term relationship with its clients; it will primarily lend to them, will trust their opinion and maintains a dialogue with them.
Targeting the top-tier shipowners	Financiers are targeting a few incumbent and top-tier shipowners.
1.2.3. Financial innovations	Financiers have started to adjust their heuristics, beliefs and tools used to make decisions to adapt to the transition of shipping and its cargo.

(continued on next page)

Table A1 (continued)

Theme/code	Description
Focus on corporate climate strategy and ESG metrics	Climate characteristics of the shipowners (climate strategy, whether it is credible or not; ESG metrics are
	developed, and data is collected) are measured and taken into account in the financiers' decisions.
Impact of the Poseidon Principles (PPs) on investment	The signing of PPs has impacted the lending behaviour of the financier.
Geolsions Knowledge build-up of demand-side risk	The financier has used resources to build an understanding of the risks and opportunities for shipping
knowledge bund-up of demand-side fisk	arising from the decarbonization of land-based sectors.
Other financial innovations	There have been attempts for financial innovations (e.g. carbon credits to finance ships; retrofit
	financing).
Risk-sharing by a larger range of actors	Traditionally, financiers would only need to share financial risk with shipowners. Financing low/zero-
	carbon ships requires this risk to be shared by a larger range of actors (charterers, public authorities e.g.
	risk.
Ship asset greenness becomes a criterion of decision	Ships' characteristics such as carbon/energy intensity, ability to retrofit and dedication to fossil-fuel
	shipping are taken into account in financiers' investment decisions.
Climate risks are incorporated into the financial risk	
analysis Unteke of groop chipping finance	The financiar has provided or is planning to provide group form of finance to shipping. This includes
optake of green shipping mance	sustainability-linked loans SBTi-aligned climate-aligned loans/bonds or inclusion of carbon intensity
	metrics on the ships' emissions in the covenants.
2. Typical behaviours	This category collects the text which supports the classification of the shipping financiers' behaviour in
	one of the ideal-typical behaviours proposed in the theoretical framework.
2.1. Creative Self-Destruction	Financiers push shipowners to invest in the incumbent technology which exacerbates the risks of
2.2. Inert	Stranged assets.
	activities as business as usual.
Full divestment is not an option	The financier rejects the option of divesting from the sector despite perceived risks.
Inertia towards LNG as a marine fuel	There is an Inertia to move away from LNG as a marine by putting faith in its drop-in fuels such as bio-
It is risky to be a first mover	methane or e-LNG.
It is risky to be a first mover	conventional. It is safer to be a follower than a first mover.
Mispricing of climate risks	Loans pricing is determined by the market competition between financiers, not by climate
	considerations.
Others are not aware of supply-side risks	Other financiers are not aware, or not as aware to supply-side risks.
Will continue financing fossil carriers	The financier intends to continue financing fossil fuel carriers.
Influence of financiers on shipowners	Financiers are necessary to shipowners and as a consequence they have a large power of influence over
	the transition to low/zero-carbon shipping.
Knowledge build-up of low/zero-carbon shipping	Knowledge build-up of low or zero-carbon shipping.
Mitigating emissions is the right thing to do	The financier justifies its action by a moral driver of the financiers to support the transition to low/zero-
Ducking the transition forequestic interacts of the	carbon shipping.
financier	the mancher reels that it will benefit from the transition to low/zero-carbon sinpping; and is merelore trying to influence the process to accelerate it
Support for LNG as a transition fuel	A financier has supported or is supporting LNG as a transition fuel.
The financier will help incumbents into doing the	Financiers will help existing incumbents/existing clients to invest in the ship assets necessary for the
transition	transition.
2.4. Redirecting Enabler	Financiers support niche actors to lead the transition to low/zero-carbon shipping.
2.5. Winding Down	requiring harder lending conditions
Replacement of traditional financiers by other	Uptake of alternative types of finance than commercial banks focusing on the incumbent shipowners to
financiers	fill the financing gap for low/zero-carbon technologies.
Fit & conform	Evidence of attempts by industry players (both niche and incumbent) and financiers to adjust the fitness
Traditional banks are retrieving from shiming for	of the shipping industry and ship assets to the expectations and needs of potential new investors.
Winding Down from demand-side risk	European Danks nave retrieved from simpling infance after 2010. Financiers are reducing their exposure/financial flows to shipping segments carrying fossil fuels
	because of the decarbonization of land-based sectors.
Winding Down from supply-side risks	Financiers are reducing their exposure/financial flows to shipping as a whole because of uncertainties
	linked to the transition to low/zero-carbon shipping.

References

Alexandridis, G., et al., 2018. A survey of shipping finance research: setting the future research agenda. Transp. Res. Part E Logist. Transp. Rev. 115 (May), 164–212. https://doi.org/10.1016/j.tre.2018.04.001.

Bach, H., et al., 2020. Implementing maritime battery-electric and hydrogen solutions: a technological innovation systems analysis. Transp. Res. Part D Transp. Environ. 87 (September), 102492.

Bach, H., Mäkitie, T., Hansen, T., Steen, M., 2021. Blending new and old in sustainability transitions: technological alignment between fossil fuels and biofuels in Norwegian coastal shipping. Energy Res. Soc. Sci. 74 (January).

Balcombe, P., Heggo, D.A., Harrison, M., 2022. Total methane and CO₂ emissions from liquefied natural gas carrier ships : the first primary measurements. Environ. Sci. Technol. 56, 9632–9640.

Baresic, D. 2020. "Sustainability transitions in the maritime transport industry : the case of LNG in Northern Europe." Doctoral thesis (Ph.D), UCL (University College London).

Bergek, A., et al., 2021. Sustainability transitions in coastal shipping: the role of regime segmentation. Transp. Res. Interdiscip. Perspect. 12 (December), 100497. Bingler, Julia, Expect the Worst, Hope for the Best: The Valuation of Climate Risks and Opportunities in Sovereign Bonds (2022). CER-ETH Working Paper 22/371,

Available at SSRN: https://ssrn.com/abstract=4076323 or https://doi.org/10.2139/ssrn.4076323.

Bolton, R., Foxon, T.J., 2015. A socio-technical perspective on low carbon investment challenges – insights for UK energy policy. Environ. Innov. Soc. Trans. 14, 165–181. https://doi.org/10.1016/j.eist.2014.07.005.

Bolton, R., Foxon, T.J., Hall, S., 2015. Energy transitions and uncertainty: creating low carbon investment opportunities in the UK electricity sector. Environ. Plan. C Govern. Policy 34 (8), 1387–1403.

Braun, V., Clarke, V., 2006. Using thematic analysis in psychology. Qualit. Res. Psychol. 3 (2), 77-101.

BRS Group, 2019. OECD workshop on trends and challenges in shipbuilding.

Bullock, S., Mason, J., Broderick, J., Larkin, A., 2020. Shipping and the Paris Climate agreement : a focus on committed emissions. BMC Energy 2 (5), 1–16.

Clarke, V., Braun, V., 2013. Successful Qualitative Research: A Practical Guide for Beginners. Sage Publications. book

Degryse, H., Goncharenko, R., Theunisz, C., Vadazs, T., 2021. When green meets green. Journal of Corporate Finance (78) 1-56. Available at SSRN: https://ssrn.com/ abstract=3724237 or https://doi.org/10.2139/ssrn.3724237.

Del Gaudio, B.L., 2018. The financing of Italian Shipping Firms. Int. Res. Econ. Financ. 2 (1), 26.

Delis, M.D., de Greiff, K., Ongena, S., 2019. Being stranded with fossil fuel reserves? Climate policy risk and the pricing of bank loans. SSRN Electron. J. 2018 (231), 1–51.

DNV GL, 2020. Energy transition outlook 2020 - maritime forecast to 2050.

Drobetz, W., Gounopoulos, D., Merikas, A., Schröder, H., 2013. Capital structure decisions of globally-listed shipping companies. Transp. Res. Part E Logist. Transp. Rev. 52, 49–76. https://doi.org/10.1016/j.tre.2012.11.008.

Faber, J., et al., 2020. Fourth IMO GHG study.

Falcone, P.M., Morone, P., Sica, E., 2018. Greening of the financial system and fuelling a sustainability transition: a discursive approach to assess landscape pressures on the Italian financial system. Technol. Forecast. Soc. Change 127, 23–37. https://doi.org/10.1016/j.techfore.2017.05.020. May 2016.

Fatica, S., Panzica, R., Rancan, M., 2021. The pricing of green bonds: are financial institutions special? J. Financ. Stab. 54, 100873.

Ferguson-Cradler, G., 2022. Corporate strategy in the anthropocene: German electricity utilities and the nuclear sudden stop. Ecol. Econ. 195 (February), 107315. Gavalas, D., Syriopoulos, T., 2015. An integrated credit rating and loan quality model: application to bank shipping finance. Mar. Policy Manag. 42 (6), 533–554. https://doi.org/10.1080/03088839.2014.904948.

Geddes, A., Schmidt, T.S., 2020. Integrating finance into the multi-level perspective : technology niche- finance regime interactions and financial policy interventions. Res. Policy 49 (6), 103985. https://doi.org/10.1016/j.respol.2020.103985.

Geddes, A., Schmidt, T.S., Steffen, B., 2018. The multiple roles of state investment banks in low-carbon energy finance: an analysis of Australia, the UK and Germany. Energy Policy 115, 158–170. August 2017.

Geddes, A., Schmid, N., Schmidt, T.S., Steffen, B., 2020. The politics of climate finance: consensus and partisanship in designing green state investment banks in the United Kingdom and Australia. Energy Res. Soc. Sci. 69, 101583 https://doi.org/10.1016/j.erss.2020.101583. May 2019.

Geels, F.W., Raven, R., 2006. Non-linearity and expectations in niche-development trajectories: ups and downs in Dutch biogas development (1973–2003). Technol. Anal. Strat. Manag. 18 (3–4), 375–392.

Geels, F.W., Schot, J., 2007. Typology of sociotechnical transition pathways. Res. Policy 36 (3), 399-417.

Geels, F.W., Kern, F., Fuchs, G., Hinderer, N., Kungl, G., Mylan, J., Neukirch, M., Wassermann, S., 2016. The enactment of socio-technical transition pathways: A reformulated typology and a comparative multi-level analysis of the German and UK low-carbon electricity transitions (1990-2014). Res. Policy 45 (4), 896–913. https://doi.org/10.1016/j.respol.2016.01.015.

Geels, F.W., 2002. Technological transitions as evolutionary reconfiguration processes: a multi-level perspective and a case-study. Res. Policy 31 (8–9), 1257–1274. Gong, S.X., Ye, H.Q., Zeng, Y.Y., 2013. Impacts of the recent financial crisis on ship financing in Hong Kong: a research note. Mar. Policy Manag. 40 (1), 1–9. Hachenberg, B., Schiereck, D., 2018. Are green bonds priced differently from conventional bonds? J. Asset Manag. 19 (6), 371–383.

Hall, S., Foxon, T.J., Bolton, R., 2016. Financing the civic energy sector: how financial institutions affect ownership models in Germany and the United Kingdom. Energy Res. Soc. Sci. 12, 5–15. https://doi.org/10.1016/j.erss.2015.11.004.

Hall, S., Foxon, T.J., Bolton, R., 2017. Investing in low-carbon transitions: energy finance as an adaptive market. Clim. Policy 17 (3), 280–298. https://doi.org/ 10.1080/14693062.2015.1094731.

Hughes, L., Downie, C., 2021. Bilateral finance organizations and stranded asset risk in coal: the case of Japan. Clim. Policy 0 (0), 1–16. https://doi.org/10.1080/ 14693062.2021.1975622.

IEA, 2021. Net zero by 2050. A roadmap for the global energy sector.

IPCC, 2022. Global warming of 1.5°C mitigation pathways compatible with 1.5°C in the context of sustainable development.

IRENA, 2019. Navigating to a renewable future: solutions for decarbonising shipping, preliminary findings.

IRENA, 2021. A pathway to decarbonise the shipping sector by 2050.

Kavussanos, MG., Tsouknidis, DA., 2014. The determinants of credit spreads changes in global shipping bonds. Transp. Res. Part E Logist. Transp. Rev. 70 (1), 55–75. https://doi.org/10.1016/j.tre.2014.06.001.

Kavussanos, MG., Tsouknidis, DA., 2016. Default risk drivers in shipping bank loans. Transp. Res. Part E Logist. Transp. Rev. 94, 71–94. https://doi.org/10.1016/j. tre.2016.07.008.

Kern, F., 2012. The development of the CCGT and the "dash for gas" in the UK power industry. Final case study report as part of work package 2 of the UKERC project: CCS – realising the potential? London.

Kungl, G., Geels, F.W., 2018. Sequence and alignment of external pressures in industry destabilisation: understanding the downfall of incumbent utilities in the German energy transition (1998–2015). Environ. Innov. Soc. Trans. 26, 78–100. January 2017.

Kungl, G., 2018. Die Grossen Stromkonzerne Und Die Energiewende. Dissertation.

Lee, K.R., Pak, M.S., 2018. Multi-criteria analysis of decision-making by international commercial banks for providing shipping loans. Mar. Policy Manag. 45 (7), 850–862. https://doi.org/10.1080/03088839.2018.1448476.

Lloyd's Register, and UMAS, 2019. Zero-emission vessels: transition pathways.

Lo, A.W., 2004. The adaptive markets hypothesis. Market efficiency from an evolutionary perspective. J. Portf. Manag. (5), 15–29.

Lucas, R., 1978. Asset prices in an exchange economy. Econometrica 46 (6), 1429-1445.

Mærsk Mc-Kinney Møller Center for Zero Carbon Shipping, 2021. We show the world it is possibe - industry transiton strategy. October.

Markard, J., Truffer, B., 2008. Technological innovation systems and the multi-level perspective: towards an integrated framework. Res. Policy 37 (4), 596–615. Markard, J., Raven, R., Truffer, B., 2012. Sustainability transitions: an emerging field of research and its prospects. Res. Policy 41 (6), 955–967.

Mitchell, J., Rehmatulla, N., 2016. Dead in the water: an analysis of industry practices and perceptions on vessel efficiency and stranded ship assets. Shipping in Changing Climates. https://discovery.ucl.ac.uk/id/eprint/1474622.

Mitroussi, K., et al., 2016. Performance drivers of shipping loans: an empirical investigation. Int. J. Prod. Econ. 171, 438–452. https://doi.org/10.1016/j. ijpe.2015.09.041.

Monk, A., Perkins, R., 2020. What explains the emergence and diffusion of green bonds? Energy Policy 145 (April), 111641. https://doi.org/10.1016/j. enpol.2020.111641.

Naidoo, Chantal (2020). Transcending the interregnum: exploring how financial systems relate to sustainability transition processes. University of Sussex. Thesis. https://hdl.handle.net/10779/uos.23477003.v1. Naidoo, C.P., 2020. Relating financial systems to sustainability transitions: challenges, demands and design features. Environ. Innov. Soc. Trans. 36 (November 2018), 270-290. https://doi.org/10.1016/j.eist.2019.10.004.

Osterkamp, P., Smith, T., Sogaard, K., 2021. Five percent zero emission fuels by 2030 needed for Paris-aligned shipping decarbonization.

Pangalos, G., 2023. Financing for a sustainable dry bulk shipping industry : what are the potential routes for financial innovation in sustainability and alternative energy in the dry bulk shipping industry? J. Risk Financ, Manag. 16 (101), 1–13.

Pavlenko, N. et al. 2020. The climate implications of using LNG as a marine fuel. www.theicct.org (February 6, 2020).

Petropoulos, T. Key Developments and Growth in Global Ship Finance. https://www.petrofin.gr/wp-content/uploads/2022/09/Petrofin-Global-Bank-Research-and-Petrofin-Index-of-Global-Ship-Finance-end2021.pdf

Pettit, S., Wells, P., Haider, J., Abouarghoub, W., 2018. Revisiting history: can shipping achieve a second socio-technical transition for carbon emissions reduction? Transp. Res. Part D Transp. Environ. 58, 292-307.

Poseidon Principles, 2020. Annual disclosure report 2020.

Sánchez, MA., 2022. A multi-level perspective on financial technology transitions. Technol. Forecast. Soc. Change 181, 1-13.

Safarzyńska, K., Frenken, K., Van Den Bergh, JC.J.M., 2012. Evolutionary theorizing and modeling of sustainability transitions. Res. Policy 41 (6), 1011-1024. Scarbrough, T. et al. 2022. Study on the readiness and availability of low- and zero-carbon technology and marine fuels. https://www.cdn.imo.org/localresources/en/

OurWork/Environment/Documents/FFTProject/Study'stechnical-proosal Ricardo DNV.pdf.

Schinas, O., Bergmann, N., 2021. The short-term cost of greening the global fleet. Sustainability 13 (16), 1–31.

Schinas, O., Ross, H.H., Rossol, T.D., 2018. Financing green ships through export credit schemes. Transp. Res. Part D Transp. Environ. 65 (September), 300-311. Science Based Targets. 2022. Science-Based target setting for the maritime transport sector. Version 1.0.

Sea Cargo Charter. "Sea Cargo Charter. About." www.seacargocharter.org (November 23, 2022). Seltzer, Lee and Starks, Laura T. and Zhu, Qifei, Climate Regulatory Risks and Corporate Bonds (2022). Nanyang Business School Research Paper No. 20-05, FRB of New York Staff Report No. 1014, Available at SSRN: https://ssrn.com/abstract=3563271 or https://doi.org/10.2139/ssrn.3563271

Seyfang, G., Gilbert-Squires, A., 2019. Move your money? Sustainability transitions in regimes and practices in the UK retail banking sector. Ecol. Econ. 156, 224–235. https://doi.org/10.1016/j.ecolecon.2018.09.014. September 2018.

ShareAction. 2019. Changing course : bank financing of the shipping industry.

Sharmina, M., McGlade, C., Gilbert, P., Larkin, A., 2017. Global energy scenarios and their implications for future shipped trade. Mar. Policy 84 (June), 12-21.

Smith, T. et al. 2019. "Scenario analysis: take-up of emissions reduction options and their impacts on emissions and costs." (July): 82. Smith, T., Bonello J.-M., and Kapur A.. 2023. How can international shipping align with 1.5°C? Focus on 1.5°C alignment in 2030. London.

Stalmokaite, I., Yliskylä-Peuralahti, J., 2019. Sustainability transitions in Baltic Sea shipping: exploring the responses of firms to regulatory changes. Sustainability 11

(7), 1-23.Stenzel, T., Frenzel, A., 2008. Regulating technological change-the strategic reactions of utility companies towards subsidy policies in the German, Spanish and UK Electricity markets. Energy Policy 36 (7), 2645-2657.

Stopford, M., 2009, Maritime Economics, Routledge, New York,

Traut, M., et al., 2018. CO₂ abatement goals for international shipping. Clim. Policy 18 (8), 1066–1075.

Tsianakidis, A., 2019. Alternative sources of finance in Greek shipping. Erasmum University Rotterdam.

Urban, M.A., Wójcik, D., 2019. Dirty banking: probing the gap in sustainable finance. Sustainability 11 (6), 1–23 (Switzerland).

Walsh, C., et al., 2019. Trade and trade-offs: shipping in changing climates. Mar. Policy 106 (February), 1–18.

Wiese Bockmann, M. 2022. "Green goes mainstream in shipping finance." Lloyd's List.

Wright, C., Nyberg, D., 2015. Climate Change, Capitalism and Corporations. Processes of Self-Destruction. Cambridge University Press. Cambridge.

Yip, A.W.H., Bocken, N.M.P., 2018. Sustainable business model archetypes for the banking industry. J. Clean. Prod. 174, 150–169. https://doi.org/10.1016/j. iclepro.2017.10.190.

Zhang, F., 2020. Leaders and followers in finance mobilization for renewable energy in Germany and China. Environ. Innov. Soc. Trans. 37, 203-224. https://doi.org/ 10.1016/j.eist.2020.08.005, August 2019,